Author's response to reviews

Title: Primary malignant chest wall tumors: analysis of 40 patients

Authors:

Reza Bagheri (bagherir@mums.ac.ir)
Seyed Ziaollah Haghi (haghiz@mums.ac.ir)
Mahmoud reza Kalantari (kalantarimr@mums.ac.ir)
Alireza Sharifian Attar (Emis@mums.ac.ir)
Maryam Salehi (salehim@mums.ac.ir)
Azadeh Tabari (azadtabari@gmail.com)
Maliheh Soudaneh (soudanehm@mums.ac.ir)

Version: 3 Date: 12 April 2014

Author's response to reviews: see over
Dear Sir / Madam

I appreciate the respectful reviewer for the comments. We have rechecked the grammar and structure of the whole manuscript and modified it. The references are checked as well and the ethical approve details are included in the "method" part of the text.

Best regards

Corresponding author: Alireza Sharifian Attar
E-mail: EMIS@mums.ac.ir
Tel: 0098-0915-315-1596
FAX: 0098-0511-8409612

Address: Assistant professor of Anesthesiology, Department of Anesthesiology, Ghaem Hospital, Faculty of medicine, Mashhad University of Medical Sciences, Mashhad, Iran